Author's response to reviews

Title: Elevation of Bombina variegata peptide 8 in mice with collagen-induced arthritis

Authors:

Daitaro Kurosaka (d_kurosaka@jikei.ac.jp)
Kentaro Noda (knoda3353@jikei.ac.jp)
Ken Yoshida (k.yoshida@jikei.ac.jp)
Kazuhiro Furuya (frykzhr@jikei.ac.jp)
Eigo Takahashi (t-815@jikei.ac.jp)
Taro Ukichi (ukichi@jikei.ac.jp)
Maimi Yanagimachi (maimy@jikei.ac.jp)
Isamu Kingetsu (i_king@jikei.ac.jp)
Saburo Saito (misaburo@jikei.ac.jp)
Akio Yamada (yamada@jikei.ac.jp)

Version: 4 Date: 27 April 2009

Author's response to reviews:

Dear Sir:

We are submitting our revised manuscript. We added Email addresses of all authors, changed Objective to Background in Abstract, removed keywords in Abstract, changed Introduction to Background, changed Materials and methods to Methods in our revised manuscript. We removed figure number and cropped space around the image in Figure.

Thank you for your consideration

Very sincerely yours.

Daitaro Kurosaka

Department of Internal Medicine, Division of Rheumatology, Jikei University school of Medicine.